Improving management of type 1 diabetes in the UK: the Dose Adjustment For Normal Eating (DAFNE) programme as a research test-bed. A mixed-method analysis of the barriers to and facilitators of successful diabetes self-management, a health economic analysis, a cluster randomised controlled trial of different models of delivery of an educational intervention and the potential of insulin pumps and additional educator input to improve outcomes

Simon Heller, 1\* Julia Lawton, 2 Stephanie Amiel, 3 Debbie Cooke, 4 Peter Mansell, 5 Alan Brennan, 6 Jackie Elliott, 1 Jonathan Boote, 6,7 Celia Emery, 1 Wendy Baird, 6 Hasan Basarir, 6 Susan Beveridge, 1 Rod Bond, 8 Mike Campbell, 6 Timothy Chater, 6 Pratik Choudhary, 3 Marie Clark, 4 Nicole de Zoysa, 9 Simon Dixon, 6 Carla Gianfrancesco, 1 David Hopkins, 9 Richard Jacques, 6 Jen Kruger, 6 Susan Moore, 10 Lindsay Oliver, 10 Tessa Peasgood, 6 David Rankin, 2 Sue Roberts, 10 Helen Rogers, 9 Carolin Taylor, 1 Praveen Thokala, 6 Gill Thompson 10 and Candice Ward 11

<sup>&</sup>lt;sup>1</sup>Sheffield Teaching Hospitals NHS Foundation Trust, Sheffield, UK

<sup>&</sup>lt;sup>2</sup>Centre for Population Health Sciences, University of Edinburgh, Edinburgh, UK

<sup>&</sup>lt;sup>3</sup>King's College London School of Medicine, London, UK

<sup>&</sup>lt;sup>4</sup>Division of Psychology, University College London, London, UK

<sup>&</sup>lt;sup>5</sup>Nottingham University Hospitals NHS Trust, Nottingham, UK

<sup>&</sup>lt;sup>6</sup>School of Health and Related Research (ScHARR), University of Sheffield, Sheffield, UK

<sup>&</sup>lt;sup>7</sup>Centre for Research into Primary and Community Care, University of Hertfordshire, Hatfield, UK

<sup>&</sup>lt;sup>8</sup>School of Psychology, University of Sussex, Brighton, UK

<sup>&</sup>lt;sup>9</sup>King's College Hospital, London, UK

<sup>&</sup>lt;sup>10</sup>Northumbria Healthcare NHS Foundation Trust, North Shields, UK

<sup>&</sup>lt;sup>11</sup>Cambridge University Hospitals NHS Foundation Trust, Cambridge, UK

<sup>\*</sup>Corresponding author

**Declared competing interests of authors:** Simon Heller has provided consultancy (for which the University of Sheffield has received payment) for Johnson & Johnson, who make blood glucose monitoring equipment, NovoNordisk and Eli Lilly who manufacture insulin. He has also given talks on behalf of NovoNordisk and Eli Lilly for which he has received payment. Gill Thompson is employed by Northumbria Healthcare NHS Foundation Trust (NHCFT) in the sole capacity of National Director – DAFNE Programme. Her salary is paid by NHCFT with funds provided by the DAFNE programme. Stephanie Amiel was an invited speaker at a pharmaceutical company-funded symposia (Eli Lilly and Company, Medtronic, Inc. and Abbott Laboratories). Her fees were paid to King's College London and her expenses were paid. Candice Ward was paid personal fees as a member of the advisory board of Cellnovo and conference speaker fees by Animas® Corporation, Cellnovo and Medtronic, Inc. Candice Ward was involved in research collaborations/projects on the use of medical devices in type 1 diabetes within the Institute of Metabolic Science; funding paid to Cambridge University Hospitals NHS Foundation Trust for service costs.

Published December 2014 DOI: 10.3310/pgfar02050

# **Plain English summary**

## The Dose Adjustment For Normal Eating (DAFNE) programme

Programme Grants for Applied Research 2014; Vol. 2: No. 5

DOI: 10.3310/pgfar02050

NIHR Journals Library www.journalslibrary.nihr.ac.uk

# **Plain English summary**

Adjustment For Normal Eating (DAFNE) trial in 2002, which showed that structured training could help adults to self-manage their glucose more successfully. Over 20,000 people have been trained and the Department of Health have stated that all individuals with diabetes in the UK should be offered structured education.

However, courses do not always work well to help people improve their blood glucose levels. We therefore investigated why some patients benefit more than others and identified barriers that would help us to develop more effective programmes.

We found that courses often fail to help participants make these skills part of their everyday lives and skills were not maintained; DAFNE graduates often needed professional support in a more structured way. We developed a course to be delivered over 1 day per week for 5 weeks and will now offer both approaches to allow more people to be trained. We developed courses to train people to use insulin pumps and to help people whose blood glucose goes too low. We also found that people with diabetes made important contributions to the research but might be more effective if involved earlier.

We now intend to develop the DAFNE programme to help people incorporate skills into their everyday lives more effectively. We will develop more effective professional support and use modern technology to help achieve this.

## **Programme Grants for Applied Research**

ISSN 2050-4322 (Print)

ISSN 2050-4330 (Online)

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: nihredit@southampton.ac.uk

The full PGfAR archive is freely available to view online at www.journalslibrary.nihr.ac.uk/pgfar. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

#### Criteria for inclusion in the Programme Grants for Applied Research journal

Reports are published in *Programme Grants for Applied Research* (PGfAR) if (1) they have resulted from work for the PGfAR programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

#### **Programme Grants for Applied Research programme**

The Programme Grants for Applied Research (PGfAR) programme, part of the National Institute for Health Research (NIHR), was set up in 2006 to produce independent research findings that will have practical application for the benefit of patients and the NHS in the relatively near future. The Programme is managed by the NIHR Central Commissioning Facility (CCF) with strategic input from the Programme Director.

The programme is a national response mode funding scheme that aims to provide evidence to improve health outcomes in England through promotion of health, prevention of ill health, and optimal disease management (including safety and quality), with particular emphasis on conditions causing significant disease burden.

For more information about the PGfAR programme please visit the website: http://www.nihr.ac.uk/funding/programme-grants-for-applied-research.htm

#### This report

The research reported in this issue of the journal was funded by PGfAR as project number RP-PG-0606-1184. The contractual start date was in October 2007. The final report began editorial review in July 2013 and was accepted for publication in April 2014. As the funder, the PGfAR programme agreed the research questions and study designs in advance with the investigators. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The PGfAR editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, CCF, NETSCC, PGfAR or the Department of Health. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the PGfAR programme or the Department of Health.

© Queen's Printer and Controller of HMSO 2014. This work was produced by Heller et al. under the terms of a commissioning contract issued by the Secretary of State for Health. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

### **Programme Grants for Applied Research Editor-in-Chief**

Professor Paul Little Professor of Primary Care Research, University of Southampton, UK

### **NIHR Journals Library Editor-in-Chief**

Professor Tom Walley Director, NIHR Evaluation, Trials and Studies and Director of the HTA Programme, UK

### **NIHR Journals Library Editors**

**Professor Ken Stein** Chair of HTA Editorial Board and Professor of Public Health, University of Exeter Medical School, UK

Professor Andree Le May Chair of NIHR Journals Library Editorial Group (EME, HS&DR, PGfAR, PHR journals)

Dr Martin Ashton-Key Consultant in Public Health Medicine/Consultant Advisor, NETSCC, UK

**Professor Matthias Beck** Chair in Public Sector Management and Subject Leader (Management Group), Queen's University Management School, Queen's University Belfast, UK

**Professor Aileen Clarke** Professor of Public Health and Health Services Research, Warwick Medical School, University of Warwick, UK

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Peter Davidson Director of NETSCC, HTA, UK

Ms Tara Lamont Scientific Advisor, NETSCC, UK

**Professor Elaine McColl** Director, Newcastle Clinical Trials Unit, Institute of Health and Society, Newcastle University, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Professor of Health Sciences Research, Faculty of Education, University of Winchester, UK

Professor John Powell Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK

**Professor James Raftery** Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Institute of Child Health, UK

**Professor Helen Snooks** Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Please visit the website for a list of members of the NIHR Journals Library Board: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: nihredit@southampton.ac.uk